Author’s response to reviews

Title: Prostate cancer outcomes in France: treatments, adverse effects and two-year mortality

Authors:

Philippe Tuppin (philippe.tuppin@cnamts.fr)
Solene Samson (Solene.samson@cnamts.fr)
Anne Fagot-Campagna (anne.fagot@cnamts.fr)
Bertrand Lukacs (bertrand.lukac@gmail.com)
Francois Alla (francois.alla@cnamts.fr)
Fred Paccaud (fred.paccaud@chuv.ch)
Jean-Christophe Thalabard (jean-christophe.thalabard@htd.aphp.fr)
Eric Vicaut (eric.vicaut@lrb.aphp.fr)
Michel Vidaud (michel.vidaud@parisdescartes.fr)
Bertrand Millat (b-millat@chu-montpellier.fr)

Version: 4  Date: 21 May 2014

Author’s response to reviews:

A suggested by the editor:

The abstract was shortened (348 words)

In Authors’ Contributions section all the authors were referred by their initials.

Sincerely yours